

\nocite{HPJava:Data_Parallel,Language_Bindings,JavaAd,Structured_SPMD,
Java_Binding_MPI,Towards_Java_SPMD,High_Level_SPMD,Considerations_HPJava,
mpiJava}

\nocite{ACHARYA96D}

\nocite{SPRINT,Adlib_classifier,CLOUDS}

